Author's response to reviews

Title: Injudicious use of Laboratory Facilities in Tertiary Care Hospitals At Rawalpindi, Pakistan: A Cross-Sectional Descriptive Study

Authors:

MUHAMMAD F MALIK (faroog_malik159@hotmail.com)
DILSHAD A KHAN (dilshad56@yahoo.com)
WAFA M ANSARI (wafamuniransari@yahoo.com)
FAROOQ A KHAN (comdtafip@gmail.com)

Version: 2 Date: 15 February 2013

Author's response to reviews:

1. The name of the Ethical Review Committee is Army Medical College, National University of Sciences and Technology Ethical Review Committee and it has been incorporated in the revised manuscript.